Author's response to reviews

Title: Promoter methylation of RASSF1A and DAPK and mutations of K-ras, p53, and EGFR in lung tumors from smokers and never-smokers

Authors:

Yang Liu (yal26@pitt.edu)
Weimin Gao (weimin.gao@TIEHH.TTU.edu)
Jill M Siegfried (siegfrie@server.pharm.pitt.edu)
Joel L Weissfeld (jwepid@pitt.edu)
James D Luketich (luketichjd@upmc.edu)
Phouthone Keohavong (pho1@pitt.edu)

Version: 2 Date: 3 May 2007

Author's response to reviews:

May 1, 2007

The BMC Editorial Team

Re: Manuscript: 1452406488134611
MS title: Promoter methylation of RASSF1A and DAPK and mutations of K-ras, p53 and EGFR in lung tumors from smokers and never-smokers.
Authors: Yang Liu, Weimin Gao, Jill M Siegfried, Joel L Weissfeld, James D Luketich and Phouthone Keohavong

Dear BMC editorial team:

We are pleased that our manuscript will be published in BMC Cancer. We are here resubmitting our revised manuscript to the journal.

The revised manuscript has now included the changes requested by the editorial team, including:

1. Author's contributions has been changed to Authors' contribution.

2. Figure legends heading have been inserted after references.

3. Tables 1 and 2 have been inserted in the manuscript after the References.

I will be happy to provide you with any further information. Please feel free to contact me.

Sincerely yours

Phouthone Keohavong, Ph.D
Associate Professor
Department of Environmental and Occupational Health
University of Pittsburgh, Pittsburgh,
100 Technology Drive, Bridgeside Point, suite 350
Pittsburgh, PA 15219-3130
Tel: (412) 624-8881
Fax: (412) 624-9361
E-mail: pho1@pitt.edu